CERES Processing Oversight Board Meeting Notes
February 4, 2015

Meeting began at 1:00 p.m. in B2102 R114.

Meeting Notes
There were no comments or questions on the notes from the last meeting.

Announcements
e CPOB Meetings will be held in B2102 R114 through May 2015.
e Denise will be sending out an email announcing a new email address for folks to send
issues to the CERES Systems team — ceres-dmt-systems@Ilists.nasa.gov..

SCCR Discussion (SCCRs that have been submitted or updated since the last meeting)

CERESIib 1054 — Contains TISA, SARB, and Instrument updates.

SARB 1055 - Synoptic SARB CERESIib updates.

GGEO 1056 — GGEO CERESIib updates.

Instantaneous SARB 1057 — Delta delivery for PGE CER5.0P3 to check for available memory
before submitting job.

Instrument 1058 — Updates to PGE CER1.4P3 to correct along-track and cross-track angle
calculations.

Instrument 1052 — Script updates to 1.1P8 to prefer to use 1.0P3 inputs.

There were no issues identified, so these SCCRs will be approved.

CATALYST Status
Weekly in-house testing continues. Expect to deliver mid to late March.

PR Tool Status
PR Database delivery planned for mid-February to add 6 new PGEs. PR Web Interface delivery
planned for mid-March.

Production Processing Status
No updates to posted report.

System Status
WebRA transition is underway for some of our Websites.
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Status of Open SCCRs

1010 — Running ValRs in PPE.

1023 — ValR data from 1010 needs to be in the DPO before operational testing can continue.
1047 — Angel is testing.

1051 — Delivery date will probably slip. Waiting for the outcome of the Feb. 11 meeting.
1054 — Will be delivered with TISA Averaging 1024.

1038 — Delivery date will probably slip. Waiting for the outcome of the Feb. 11 meeting.
1024 — Investigating shortwave inconsistencies. Delivery next week at the earliest.

1017 — Will be delivered after SCCR 1024.

1025 — Code is running on both platforms. Working on comparison software.

919 — Angel is testing.

1049 — No updates.

1050 — Science development continues.

1052 — Delivery date will probably slip. Waiting for the outcome of the Feb. 11 meeting.
1057 — Delivered 2/4/15. CM testing will be done when system load is high.

1058 - Target delivery date is 2/27/15.

Meeting ended at 1:46 p. m.

Page 2 of 2



